Author's response to reviews

Title: Fertility in four regions spanning large contrasts in serum levels of widespread persistent organochlorines: a cross-sectional study

Authors:

Gunnar Toft (gutof@as.aaa.dk)
Anna Axmon (anna.axmon@med.lu.se)
Aleksander Giwercman (aleksander.giwercman@kir.mas.lu.se)
AneMarie Thulstrup (anemt@as.aaa.dk)
Anna Rignell-Hydbom (anna.rignell-hydbom@ymed.lu.se)
Henning Sloth Pedersen (hsp@natur.gl)
Jan Ludwicki (K.LUDWICKI@PZH.GOV.PL)
Valentina Zvyesday (dimusic@ic.kharkov.ua)
Alex Zhinchuk (dimusic@ic.kharkov.ua)
Marcello Spano (spanomrc@casaccia.enea.it)
GianCarlo Manicardi (manigi04@mail.unimo.it)
Eva Bonefeld-Jorgensen (ebj@mil.au.dk)
Lars Hagmar (lars.hagmar@ymed.lu.se)
Jens Peter Bonde (jpbon@as.aaa.dk)

Version: 6 Date: 3 November 2005

Author's response to reviews:

Dear editors:

We have updated the manuscript according to the additional requests with respect to formatting:

The details in the letter has been corrected. All suggested changes in the manuscript has been accepted. Two sentences have been rephrased. The format of Table titles and footnotes have been made uniform. Finally, the Tables have been resized once more to fit between the margins.

Looking forward to your response - and will be happy to respond to additional requests,

yours sincerely with kind regards,

Jens Peter Bonde